Author’s response to reviews

Title: A rare presentation of pheochromocytoma in pregnancy: a case report

Authors:

seyedeh noushin ghalandarpoor attar (noushinattar64@gmail.com)
sedigheh borna (s_borna@hotmail.com)
seyedeh mojgan ghalandarpoorattar (dr.ghalandarpoor@yahoo.com)
fahimeh ghotbizadeh (fghotbi@yahoo.com)

Version: 1 Date: 11 Dec 2017

Author’s response to reviews:

Reviewer #1

4. Does the article report the following information? Where information is missing, please specify.

THANKS FOR YOUR COMMENT DEAR REVIEWER.

i wish to say the adrenal mass was not diagnosed earlier because in fetal anomaly scan just fetus is assessed and maternal organs exept uterus and adnexes are not included .our patint suffered flank pain and this was the clue to assess maternal kidneys and subsequent adrenal mass diagnosis

5. Is the interpretation (discussion and conclusion) well balanced and supported by the case presented?

discussion was changed

8. Does the case represent a useful contribution to the medical literature?

Comments:

Not much unless they improve the discussion.

discussion was changed